Author's response to reviews

Title: Oxytocin and cholecystokinin secretion in women with colectomy

Authors:

Bodil Ohlsson (bodil.ohlsson@kir.lu.se)
Jens F Rehfeld (JensFRHO3561Rehfeld@rh.dk)
Mary L Forsling (bodil.ohlsson@kir.lu.se)

Version: 3 Date: 22 September 2004

Author's response to reviews: see over
Answers to the Editor

1. I was not able to crop the figures. I have asked for help with the first one, and hope that next one will be correct. If not, I hope you can help me with that too, or tell me, then I ask for help.
2. The short title has been removed from the title page.
3. We have changed the spelling of clearance in the abstract.
4. The E mail addresses are included now.
5. The key words are removed.
6. Introduction has changed name to Background.
7. Subjects & Methods are now Methods.
8. The Table has been changed and the reviewers comments are put in; The number of subjects and plasma.
9. One section is called conclusion.
10. Abbreviations has been moved.
11. A list of what each author has done is included in front of Acknowledgements, where we also declare that we have no competing interests.
12. The reference list has been changed, and squared brackets are present in the text.

We now hope that everything is OK. If not, please let us know so we can change.

Sincerely

Bodil Ohlsson